Author’s response to reviews

Title: Abdominal wall hemorrhage after intravenous thrombolysis for acute ischemic stroke

Authors:

Se-A An (ansea@chamc.co.kr)
Jinkwon Kim (kimjinkwon@chamc.co.kr)
Sang Heum Kim (shss96@cha.ac.kr)
Won Chan Kim (wckim@cha.ac.kr)

Version: 2 Date: 31 August 2012

Author’s response to reviews:

August 28, 2012

Deesha Majithia
Executive Editor
BMC Neurology

Dear Editor:

I and my coauthors would like to submit the manuscript entitled “Abdominal wall hemorrhage after intravenous thrombolysis for acute ischemic stroke” for publication as a Case Report in BMC Neurology.

We report abdominal wall hematoma as a rather unique complication of thrombolysis for acute ischemic stroke. Although bleeding is known to be a complication of this procedure, reports of extracranial bleeding are very rare and, from what we understand, this would be the first report on abdominal bleeding. The findings stress on the importance of considering such a bleeding in cases of patients who complain of abdominal pain/discomfort after undergoing intravenous thrombolysis for acute ischemic stroke.

We believe that the findings of this study are relevant to the scope of your journal and will be of interest to its readership.

This manuscript has not been published or presented elsewhere in part or in entirety, and is not under consideration by another journal. The patient has provided her consent for publication of this case report. All the authors have approved the manuscript and agree with its submission to your esteemed journal. There are no conflicts of interest to declare.
I look forward to hearing from you at your earliest convenience.

Sincerely,

Won Chan Kim, M.D.
Department of Neurology
CHA Bundang Medical Center
CHA University College of Medicine
351 Yatap-dong, Bundang-gu,
Seongnam 463-712, Republic of Korea
Tel: +82-31-780-5482
Fax: +82-31-780-5269
E-mail: wckim@cha.ac.kr